Author's response to reviews

Title: Can care of elderly be measured? Psychometric testing and modification of the Time in Care instrument for measurement of care needs in nursing homes

Authors:

Kajsa BE Thorsell (Kajsa.Thorsell@hassleholm.se)
Berit M Nordström (Berit.Nordstrom@med.lu.se)
Per Nyberg (Per.Nyberg@med.lu.se)
Bengt V Sivberg (Bengt.Sivberg@med.lu.se)

Version: 5 Date: 17 September 2008

Author's response to reviews:

Dear Editor BMC Geriatric,

We have changed all four tables into requested journal format, and done the lay-out changes we proposed in an earlier mail. Also, some misspellings have been corrected.

Bengt Sivberg